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OCO-2 Status Summary

Å Observatory Status: Nominal

ïCollecting science data in gyroless mode with the optimized science operating scheme

ïDrag Make-up Maneuver #23 executed today (9 Jan 2020)

Å Instrument Status:  Nominal é again

ïO2 A-band color slice updating issue resolved. 

ïNext Decon scheduled for 21- 28 January 2020

Å Science Status: Nominal

ïBuild 10 (B10) ï14 months of data were processed using the Build 10 algorithm

ÅAbout half on Pleiades and half Amazon Cloud (while OCO-2 computers were relocated)

ÅThese data will be used to illustrate the B10 product in the Senior Review proposal

ÅThe reprocessing campaign will continue through July 

ï ACOS GOSAT v9 ïAlmost There

ÅChris OôDell has completed a preliminary filtering and bias correction process

ïACOS v9 product covers a 10-year period 

ïIncludes both High and Medium gain land and High gain ocean 

» For v7.3, no bias correction was created for Land Gain M (bright desert regions)
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FPA Ice Buildup Trending

Å Throughput now at 91.5%

Å Rate of ice buildup 

continues to decrease

Å Next DECON scheduled for 

the week of 21-28 January

ïNo science data will be 

collected during that week

~91.5% 

Lars Chapsky
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The A-Band Color Slice Anomaly

Å An A-band anomaly was first detected by 

Richard Leeôs housekeeping trend data

ï In orbit 29292 on 3 January 2020, the A-

band radiance data, derived from color 

slices, stopped changing

Å This was initially interpreted as an issue 

with the trending software, but later 

traced to the housekeeping data that is 

routinely downlinked by the instrument

ïThe instrument is updating the headers of 

the ABO2 optical housekeeping packet 

but color slice data in the body of the 

packets is not being updated

Å An anomaly investigation was opened 

and is ongoingRichard Lee
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The A-Band Color Slice Anomaly is Resolved

The OCO-2 Ops team uplinked a command that power cycled the O2 A-band data processor. This command 

appears to have resolved the A-band Color Slice updating issue.

Reset
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OCO-2 A-Band Color Slice Data Anomaly

Å An anomaly investigation was initiated to identify root cause and corrective action needed

ïNature of anomaly - Color slice data are not being updated in the Housekeeping data stream

ïTime: January 3, at around 18:11 Z Orbit 29292 

ï Location: Over Brazil, within the South Atlantic Anomaly, suggesting that this may be a cosmic ray hit

Å Corrective actions

ïThe OCO-2 Ops team uplinked a command that power cycled the O2 A-band data processor. This 

command appears to have resolved the A-band Color Slice updating issue.

Å Impacts to Science

ïThe range of orbits affected: 29292 (18:11 UTC on 3 Jan 2020) to 29368 (~22:30 UTC on 8 Jan 2020)

ïThe ñde-clockingò (i.e. FPA rotation) correction algorithm requires color slice data. 

ïData processed through the regular pipeline during this 5.25 day period is not likely to be useful for science 

ïWe may be able to salvage some of this data by reprocessing with de-clocking turned off

Å The system has returned to nominal data collection
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Preliminary ACOS GOSAT v9 Filter Plots

Land Gain H vs Models Land Gain M vs Models Ocean Gain H vs Models

Composite vs Models

Land Gain H vs Models Ocean Gain H vs ModelsChris OôDell
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ÅMost of the month of November was spend with PMA Cal

ïThis is an activity to collect images with the internal camera that are compared to Landsat images

ïData used to develop a pointing correction table

ÅDecember ïtesting and implementation

ïwe used past data to test pointing correction table. Find that we have agreement with coastlines 

to 500-1000m

ïThis was uploaded to the instrument on Dec 17th

ïCan see the improved pointing in SAMs taken since then

ÅNew version of data processing software went operational on Dec 17th. This has the 

pointing correction and other fixes

OCO-3 Status Update (1/7/2020 pg 1)
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Å January data release

ïL1b data is being produced and sent to the DAAC

ïwe are completing data checks and documentation

ïexpect data released to the public in a couple of weeks

Å Looking ahead

ïcontinue improvement of pointing corrections

ïdive into L2 data, have a small forward stream, and will reprocesses useful data from the past 

(targets, some SAMS, some ND/GL periods)

Åwill use same approaches as OCO-2. CSU team on the critical path!

ïtune up a few aspects of data collection

Åoptimize glint signals

ÅSAMs that are more spaced out or not square

Åshorten some targets 

ïimprove tools for planning, tracking what we collected, and sharing info

OCO-3 Status Update (1/7/2020 pg 2)
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Å L2 delivery 90 days after L1b (March/April)

Å focused alg/cal/val meeting at Caltech March 24/25/26

Åfull STM in Boulder Oct 20/21/22 (may be half day moreé.)

Future Milestones
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Geolocation performance with calibration of pointing system

1st set of 

operational 

data

geolocation 

optimization 

offline

current 

operational 

data

Figure provided by R. Nelson, JPL/Caltech


